Author’s response to reviews

Title: Lymphocyte apoptosis in children with central nervous system tuberculosis: a case control study

Authors:

Paola Di Carlo (paola.dicarlo@unipa.it)
Alessandra Casuccio (alessandra.casuccio@unipa.it)
Amelia Romano (ameliaromano@libero.it)
Daria Spicola (dariaspicola@yahoo.it)
Lucina Titone (lucina.titone@unipa.it)
Nadia Caccamo (nadia.caccamo@unipa.it)
Francesco Dieli (francesco.diel@unipa.it)
Caterina Mammina (caterina.mammina@unipa.it)
Elisabetta Pace (pace@ibim.cnr.it)
Mark Gjomarkaj (gjomarkaj@ibim.cnr.it)
Mario Melis (mario.raphael.melis@ibim.cnr.it)
Manlio Tolomeo (mtolomeo@hotmail.com)

Version: 5 Date: 21 November 2011

Author’s response to reviews:

Dear Sir,

Please, find enclosed a revised version of the manuscript Paper number MS: 1089850365582193

Title: Lymphocyte apoptosis in children with central nervous system tuberculosis

Authors: Paola Di Carlo, Alessandra Casuccio, Amelia Romano, Daria Spicola, Lucina Titone, Nadia Caccamo, Francesco Dieli, Caterina Mammina, Elisabetta Pace, Mark Gjomarkaj, Mario Melis and Manlio Tolomeo

The Following changes have been made:

• AUTHOR LIST: the list of authors in the manuscript are written exactly as they are in the submission system, both in style and order
  Paola Di Carlo 1, Alessandra Casuccio 2, Amelia Romano 3, Daria Spicola 1, Lucina Titone 1, Nadia Caccamo 4, Francesco Dieli 4, Caterina Mammina 1, Elisabetta Pace 5, Mark Gjomarkaj 5, Mario Melis 5 AND Manlio Tolomeo 1

• AFFILIATIONS: the following affiliation is written exactly as it is in Methods section of Manuscript
  3. Pediatric Infectious Diseases, G. Di Cristina Children’s Hospital, ARNAS Civico, P.zza Montalto, 8, Palermo I-90157, Italy

• Corresponding author: the affiliation is written exactly as it is in AFFILIATIONS
• Paola Di Carlo, MD, e-mail: paola.dicarlo@unipa.it, Address: Department of Sciences for Health Promotion, University of Palermo, Via del Vespro 133, Palermo I-90127, Italy. Tel. : + 39 091 6554057, Fax : + 39 091 6554060

Sincerely,
Paola Di Carlo